Author’s response to reviews

Title: Black and White Men Younger than 50 Years of Age Demonstrate Similar Outcomes After Radical Prostatectomy

Authors:

Kelvin A. Moses (kelvin.a.moses@vanderbilt.edu)
Ling Y. Chen (chenl3@mskcc.org)
Daniel D. Sjoberg (sjobergd@mskcc.org)
Melanie Bernstein (bernstem@mskcc.org)
Karim A. Touijer (touijera@mskcc.org)

Version: 3
Date: 20 August 2014

Author’s response to reviews:

Thank you for the opportunity to revise our manuscript. We have made the corrections to the methods section as requested. The first paragraph now includes the name of the IRB and the reason for the waiver of patient consent.

Thank you and please let us know of any questions or further information needed.

Sincerely,

Kelvin Moses